Author’s response to reviews

Title: Factors associated with good self-rated health of non-disabled elderly living alone in Japan: a cross-sectional study

Authors:

Wei Sun (sunwei80@yahoo.com)
Misuzu Watanabe (misuzuw@art.osaka-med.ac.jp)
Yoshimi Tanimoto (hyg022@art.osaka-med.ac.jp)
Takahiro Shibutani (hyg024@art.osaka-med.ac.jp)
Rei Kono (hyg031@art.osaka-med.ac.jp)
Masahisa Saito (saito-m@art.osaka-med.ac.jp)
Kan Usuda (hyg032@art.osaka-med.ac.jp)
Koichi Kono (hyg007@art.osaka-med.ac.jp)

Version: 4 Date: 12 October 2007

Author’s response to reviews:

Dear The BioMed Central Editorial Production Team:

Thanks a lot for the formatting requirements on my paper (MS: 2434520851296770). I changed my paper as required.

Now, I would like to give a point-by-point description to the requirements.

Q1. References - The reference list should contain all authors’ names, if there are 30 or below. The term ‘et al.’ should not be used (ref 6-8, 23, 24, 26-28, 30, 35).

Answer: I checked all my reference papers again and gave all authors’ names.

Q2. tables- please change ‘Table1’ to ‘Table 1’ and ‘table2’ to ‘Table 2’.

Answer: I changed them as you advised.

Q3. Table 2 column headings - This table currently runs over two pages and the column headings have been repeated. In order to ensure that the table is correct in the final PDF, please can you ensure that it is made into one table. This can run on to the next page, but please remember to remove the second set of column headings.

Answer: I changed Table 2 into just one table that runs on to the next paper without the second set of column heading.

These changes are made as advised. If there are more formatting requirements, we are looking to getting your email again. Thanks a lot!

Best wishes,

Sincerely
Wei Sun